Author's response to reviews

Title: Bladder irrigation and urothelium disruption: a reminder apropos of a case of fatal fluid absorption.

Authors:

Marco Di Paolo (marcodipaolo@med.unipi.it)
Valentina Bugelli (vale.buge@hotmail.it)
Alessandro Di Luca (alessandro.diluca@uniroma1.it)
Emanuela Turillazzi (emanuela_turillazzi@inwind.it)

Version: 3
Date: 13 March 2014

Author's response to reviews:

Dear Editor, I'm submitting a new revised version of the manuscript. A native speaker revised the paper.

I have a concern about the consent. This is a court case in which the autopsy was arranged by the local prosecutor.

The patient's relatives themselves, at the time they made a complaint to the prosecutor, provided all the medical documentation and agreed to carry out the autopsy. The forensic pathologist in Italy has no direct contact with deceased's relatives. I can only provide a consent by the prosecutor’s office.

Many thanks for your kind collaboration

Best regards

Emanuela Turillazzi, MD, phd.